Author’s response to reviews

Title: Demographic, Psychosocial and Clinical Factors Associated with Postpartum Depression in Kenyan Women

Authors:

Linnet Ongeri (linongeri@gmail.com)
Valentine Wanga (wangav@uw.edu)
Phelgona Otieno (phelgona@gmail.com)
Jane Mbui (Jane.Mbui@usamru-k.org)
Elizabeth Juma (jumaelizabeth@yahoo.com)
Ann VanderStoep (annv@uw.edu)
Muthoni Mathai (muthonimathai@gmail.com)

Version: 1 Date: 02 Dec 2017

Author’s response to reviews:

Dear Editor,

We have revised the following in the manuscript submission

1. We have included email addresses of all co-authors in the title page of the manuscript.

2. We confirm in the ethics statement that that a written consent form was provided to the participants.

3. A statement regarding Ethics and Consent in the methods section has been included.
4. Figures from our manuscript have been removed and uploaded as individual figure files as stated in your guidelines.

5. Duplicate Table files have been removed.